Author's response to reviews

Title: GPs’ decisions on drug treatment for patients with high cholesterol values: A think-aloud study

Authors:

Lars G Backlund (lars.backlund@klinvet.ki.se)
Ylva Skaner (ylva.skaner@klinvet.ki.se)
Henry Montgomery (hmy@psychology.su.se)
Johan Bring (johan.bring@statisticon.se)
Lars-Erik Strender (lars-erik.strender@klinvet.ki.se)

Version: 3 Date: 8 December 2004

Author's response to reviews:

The following changes have been made:
1. The article published in BMC Family Practice has been discussed by two additions under Discussion. The first is the paragraph on page 21, begins with "The use of life style factors.." and ends with "..strategies for intervention." The second is the sentence that ends the first paragraph on page 23 and begins with "However,...".

The reference list has been completed with the BMC Family Practice article. We have also exchanged the references to the Swedish and European guidelines to the most recent ones (which does not affect the content of the article in other ways).

The Abstract has been added to page 2. The Competing interest section has been changed according to standard. We agree that a link is created to BMC Cardiovascular Disorder.

The changes regarding Figures and Tables has been made according to recommendations.

Finally, we are sending a letter today with approval for publication from the review board.

With best regards

Lars Backlund